Author’s response to reviews

Title: Association between renal function and bone mineral density in healthy postmenopausal Chinese women

Authors:

Shuang Li (ls330013@csu.edu.cn)
Junkun Zhan (zhanjunkun@csu.edu.cn)
Yanjiao Wang (917637923@qq.com)
Yi Wang (13874857512@139.com)
Jieyu He (9953726@qq.com)
Wu Huang (huangwu_2008@yeah.net)
Zhifeng Sheng (ls330013@163.com)
Youshuo Liu (liuyoushuo@csu.edu.cn)

Version: 1 Date: 09 Sep 2019

Author’s response to reviews:

Dear Editors:
RE: BEND-D-19-00337

Thank you very much for your letter and advice. Following the suggestions of the editors, we have revised the manuscript. We herewith submit the revised paper for re-examination. We hope that the revised version of the manuscript is now acceptable for your journal.

We look forward to hearing from you soon.
With best wishes,

Yours sincerely,
Youshuo Liu